

THE UNIVERSITY of EDINBURGH

Edinburgh Research Explorer

Investigating a strategy for quantifying schistosome infection levels in preschool-aged children using prevalence data from school-aged children

Citation for published version:

Lim, R, Woolhouse, M, Mduluza, T, Chase-Topping, M, Osakunor, D, Chitsulo, L & Mutapi, F 2020, 'Investigating a strategy for quantifying schistosome infection levels in preschool-aged children using prevalence data from school-aged children', *PLoS Neglected Tropical Diseases*, vol. 14, no. 10, e0008650. https://doi.org/10.1371/journal.pntd.0008650 O

Digital Object Identifier (DOI):

10.1371/journal.pntd.0008650 O

Link:

Link to publication record in Edinburgh Research Explorer

Document Version: Publisher's PDF, also known as Version of record

Published In: PLoS Neglected Tropical Diseases

General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy

The University of Édinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.





Citation: Lim RM, Woolhouse MEJ, Mduluza T, Chase-Topping M, Osakunor DNM, Chitsulo L, et al. (2020) Investigating a strategy for quantifying schistosome infection levels in preschool-aged children using prevalence data from school-aged children. PLoS Negl Trop Dis 14(10): e0008650. https://doi.org/10.1371/journal.pntd.0008650

Editor: Uwem Friday Ekpo, Federal University of Agriculture Abeokuta, NIGERIA

Received: February 3, 2020

Accepted: July 27, 2020

Published: October 1, 2020

Copyright: © 2020 Lim et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: Our paediatric schistosomiasis project is funded by the Thrasher Research Fund 12440 and the Wellcome Trust 108061/Z/15/Z. This research was also commissioned by the National Institute for Health Research (NIHR) Global Health Research programme (16/136/33) using UK AID from the UK Government. The views expressed in **RESEARCH ARTICLE**

Investigating a strategy for quantifying schistosome infection levels in preschoolaged children using prevalence data from school-aged children

Rivka M. Lim¹*, Mark E. J. Woolhouse^{2,3}, Takafira Mduluza⁴, Margo Chase-Topping^{2,5}, Derick N. M. Osakunor¹, Lester Chitsulo⁶, Francisca Mutapi^{1,3}

 Institute of Immunology & Infection Research, University of Edinburgh, Ashworth Laboratories, Edinburgh, United Kingdom, 2 Usher Institute, University of Edinburgh, Ashworth Laboratories, Edinburgh, United Kingdom, 3 NIHR Global Health Research Unit Tackling Infections to Benefit Africa (TIBA) at the University of Edinburgh, Ashworth Laboratories, Edinburgh, United Kingdom, 4 Department of Biochemistry, University of Zimbabwe, Mount Pleasant, Harare, Zimbabwe, 5 Roslin Institute, Easter Bush, Midlothian, United Kingdom, 6 Lilongwe, Malawi

* rivka.lim@ed.ac.uk

Abstract

In 2012, the World Health Organisation (WHO) set out a roadmap for eliminating schistosomiasis as a public health problem by 2025. To achieve this target, preschool-aged children (PSAC; aged 6 years and below) will need to be included in schistosomiasis treatment programmes. As the global community discusses the tools and approaches for treating this group, one of the main questions that remains unanswered is how to quantify infection in this age group to inform treatment strategies. The aim of this study was thus to determine whether a relationship exists between levels of schistosome infection in PSAC and schoolaged children (SAC), that can be used to determine unknown schistosome infection prevalence levels in PSAC. A systematic search of publications reporting schistosomiasis prevalence in African PSAC and SAC was conducted. The search strategy was formulated using the PRISMA guidelines and SPIDER search strategy tool. The published data was subjected to regression analysis to determine if a relationship exists between infection levels in PSAC and SAC. The interaction between SAC and community treatment history was also entered in the regression model to determine if treatment history significantly affected the relationship between PSAC and SAC prevalence. The results showed that a significant positive relationship exists between infection prevalence levels in PSAC and SAC for Schistosoma mansoni (r = 0.812, df (88, 1), p = <0.0001) and S. haematobium (r = 0.786, df (53, 1), p = <0.0001). The relationship was still significant after allowing for diagnostic method, treatment history, and the African sub-region where the study was conducted (S. mansoni: F = 25.63, df (88, 9), p = <0.0001; S. haematobium: F = 10.20, df (53, 10), p = <0.0001). Using the regression equation for PSAC and SAC prevalence, over 90% of the PSAC prevalence studies were placed in the correct WHO classifications category based on the SAC levels, regardless of treatment history. The study indicated that schistosome prevalence in SAC can be extended as a proxy for infection levels in PSAC, extending on its current use in the

this publication are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

adult population. SAC prevalence data could identify where there is a need to accelerate and facilitate the treatment of PSAC for schistosomiasis in Africa.

Author summary

Preschool-aged children (PSAC), i.e. aged ≤ 6 years, are not included in preventative chemotherapy programmes, currently advocated by the WHO for the treatment of schistosomiasis. This is due to the lack of a paediatric formulation of the drug of choice, praziquantel, and the current guidelines which requires diagnosis before treatment. As the global community prepares for the deployment of a new paediatric formulation of praziquantel, there is a need to find a strategy to quantify infection in this age group. In schistosome endemic areas, infection levels in school-aged children (SAC) are already used to inform infection levels in the community. Thus, we investigated the relationship between SAC and PSAC schistosome prevalence levels within the same community, to determine if data from SAC could be used to predict infection levels in PSAC. Our results show that PSAC prevalence levels are significantly correlated with SAC prevalence levels. Our findings are applicable to communities that have received preventative chemotherapy in SAC or in SAC and adults, as well as those that have not received any. Our study indicates that it is possible to extrapolate PSAC prevalence levels from SAC prevalence, and to make a treatment decision on that basis.

Introduction

Schistosomiasis is a widespread parasitic disease found in tropical and subtropical areas [1]. The World Health Organization (WHO) estimated that at least 206 million people worldwide required treatment for schistosomiasis in 2016, with at least 91% living in Africa [2]. The most widely used treatment for schistosomiasis is the antihelminthic drug of choice, praziquantel (PZQ), which is both safe and efficacious against adult worms [3]. Currently, schistosomiasis is controlled through preventative chemotherapy, targeting school-aged children (SAC) who are treated with PZQ through mass drug administration (MDA). The frequency of treatment follows guidelines from the WHO, which are based on the schistosome endemicity of the area rather than individual infection status [4]. The endemicity of the area is determined by quantifying schistosome infection prevalence in SAC, following sampling of a group of the children. The reason SAC are used for determining community schistosome endemicity and are the primary target of MDA is that they have been shown to have the highest prevalence of infection and are easily accessible in schools [5, 6].

As of November 2018, 29 of the 41 African countries requiring preventive chemotherapy have implemented MDA preventative chemotherapy in SAC; approximately 75 million children have already received treatment [7]. However, none of these programmes include preschool-aged children (PSAC; aged 6 years and below). In endemic areas, children can be infected as early as 1 year old, and the burden of infection as well as disease morbidity increases with age [8, 9]. Several authors have highlighted the gap in treatment strategies, created by the exclusion of PSAC (reviewed by Stothard *et al.* 2013) [10]. This creates a health inequity for approximately 50 million African PSAC exposed to schistosomiasis [11].

In order to eliminate schistosomiasis as a public health problem by 2025, based on the WHO roadmap on neglected tropical diseases (NTDs) [12], there is a need to include PSAC in

treatment programmes. The WHO have responded by recommending the treatment of PSAC [13], and also calling for the development of a paediatric formulation of PZQ, suitable for treating PSAC. There is now a paediatric formulation of PZQ under development (currently undergoing Phase III clinical trials) targeting children aged 3 months to 6 years. Thus, in preparation for inclusion of PSAC in treatment programmes, either as part of MDA or treatment upon diagnosis, there is a need to develop a strategy for quantifying infection in this age group.

Unlike SAC who are readily accessible in schools, PSAC are not often as accessible, and conducting mapping exercises to quantify their infection levels can be challenging and costly. Therefore, we aimed to determine if SAC infection prevalence data could be used to determine infection prevalence in PSAC. In this study, we investigated the relationship between schistosome prevalence levels in SAC and PSAC, to determine if SAC infection levels in Africa can be used as a predictor of the prevalence levels in PSAC from the same community [13, 14].

Methods

This study focused on published manuscripts reporting schistosome infection prevalence, derived from parasitological detection of schistosome eggs excreted in urine or stool. A systematic search of publications that reported schistosomiasis prevalence in PSAC and SAC in Africa was carried out. The search strategy was formulated using the PRISMA guideline (Preferred Reporting Items for Systematic Reviews and Meta-Analysis) [15], and the SPIDER search strategy tool (Sample, Phenomenon of Interest, Design, Evaluation, Research type) [16] in order to formulate a search question (see Table 1).

Published studies were identified using three online databases: PubMed (conception to 30th October 2018), Embase (1980–2018, week 44) and Web of Science (1983 – 30th October 2018). The results were filtered to show only those published in English.

Criteria for inclusion and exclusion of studies

Results from the literature searches were consolidated and extracted using EndNote referencing software (X8.01). Endnote and a Python script were used to match and remove duplicate titles.

Eligibility for inclusion was based on the following criteria: i) the survey was carried out in Africa, ii) it was published/written in English, iii) it contained prevalence data for PSAC, SAC

Sample	Children (preschool and school age)				
Phenomenon of interest	Schistosomiaisis infection, either S. haematobium or S. mansoni				
Design	Cross-section, survey				
Evaluation Egg count in urine or stool					
Research type Not included in search criteria but reviews and meta-analysis were removed during screening					
Terms used in search					
S	Children OR Preschool OR Pre - school OR Infant OR Infants OR PSAC OR SAC				
	AND				
PI	Schisto* OR Bilharzia				
	AND				
D	Cross section OR Cross-section OR Cross-sectional OR Survey OR Prevalence				
	AND				
E	Urine OR Stool OR Katz OR egg OR eggs				
R	N/A				

Table 1. Search terms created using the SPIDER search strategy.

https://doi.org/10.1371/journal.pntd.0008650.t001

or both, iv) the sample size was clearly stated, v) the sample size for each age group was above 10 people, vi) parasites identified were either *S. mansoni* or *S. haematobium*, vii) diagnosis was via egg count in stool or urine samples, viii) the age ranges were clearly defined, and ix) PSAC age range was between 0 up to 6 years and SAC age range was between 6–16 years.

The studies were further assessed using the following criteria: recruitment of participants was random and non-biased, and sample collection and diagnostics was clearly explained and carried out according to standard protocols [6]. Any studies not meeting these criteria were excluded from the analyses. Data were entered based on prevalence of schistosomiasis per water source/village, therefore one citation could yield multiple datasets.

Quality assessment

The quality of each study included in the analysis was assessed using a graded scale. Each study was scored either 1 (yes) or 0 (no) against multiple criteria. There were 7 criteria–i) population sampling methods described; ii) results stratified by gender; iii) a description of laboratory diagnostics included; iv) multiple samples taken on different days; v) multiple tests of a single sample carried out (*S. mansoni* only); vi) water source and/or the sanitary conditions described; and vii) statistical analysis described in the methods. The scores were added together and the overall score per article graded as follows: high 6–7, medium 3–5, and low 0–2 for *S. mansoni*, and for *S. haematobium*, high 5–6, medium 3–4 and low 0–2. If any of the studies scored low on quality assessment, the analysis was conducted with and without the study, to determine if it had an effect on the overall results.

Study variables

Study variables extracted from the publications were: sample size, age range, egg detection method used for schistosome diagnosis, schistosome treatment history of the study communities, and the antihelminthic drug used for the treatment. The studies used different age ranges for defining PSAC and SAC. For this study PSAC was defined as ages 0-6 years and SAC were defined as 6 years up to 16 years of age. Techniques for detecting eggs in stool for S. mansoni diagnosis were i) formol-ether sedimentation [17], ii) the Bell Method [18], and iii) Kato-Katz thick smear [19]. For S. haematobium, egg techniques reported were urine sedimentation and filtration methods [20]. Studies were included in the analysis providing the paired PSAC and SAC infection were diagnosed using the same technique. History of schistosome treatment of the community or the study participants was recorded and included in the statistical analyses; this information was available at a community/group level but not individual participant level. Four categories of treatment history were recorded; 1) No treatment taken place prior to study (i.e. no recorded mass treatment in the past 50 years), 2) Treatment currently underway, 3) Treatment has been recorded within the past 5 years, and 4) No information available about treatment history. If the study did not state whether the community had undergone previous MDA, the WHO Preventative Chemotherapy Treatment (PCT) databank was used to assess whether treatment strategies were in place in this country or had taken place in the past 5 years [21].

Data extraction

All eligible articles were read in full. The following information was extracted and tabulated (Microsoft Excel); first author name, year of publication, dates of survey, country, African subregion, sample size, age range, male to female ratio, overall prevalence of infection, infection intensity, male and female infection prevalence, standard deviation, standard error of the mean and 95% confidence intervals, previous schistosome treatment history and method used for collecting the eggs for schistosomiasis diagnosis.

If the data were presented in smaller age ranges, for example, one of the studies specified groups were 6–10 years and 10–16 years, the sample sizes were added together, and from this combined group an average prevalence was calculated. When the age groups exceeded the limits defined by this study e.g. above 6 years old for PSAC or above 16 years old for SAC, the study was removed from the analysis. Furthermore, if the SAC group was partitioned into two, and the older group exceeded the upper limit of the grouping, only the younger group within the SAC was included. Where the published data were presented in graphs, Data-Thief software (Data Thief III), a highly accurate data extraction tool, was used to extract the exact numerical data [22].

Two categories of data were identified, Category 1 in which both PSAC and SAC data come from the same study, and Category 2 in which the data were from two separate studies. Category 2 studies were matched using the following additional criteria: the surveys were carried out within one year of each other, the population used the same water source, and the parasitology diagnostic techniques were the same.

Statistical analysis

Data analysis was carried out using MINITAB (version 18). As the infection data for both SAC and PSAC were provided as prevalences (recorded as percentage data), both variables were transformed using the arcsine square root transformation to satisfy the assumptions for parametric tests, and to avoid resultant negative prevalence values post-transformation [23]. We determined whether or not data satisfied the parametric assumptions through analysing the residuals (e.g. normality plots and histogram). We also validated the regression model by analysing the goodness of fit of the regression and analysing whether the regression residuals are random.

To determine the relationship between PSAC and SAC, Pearson's correlation and linear regression without any confounding factors were used. Thereafter, the relationship was analysed through linear regression, allowing for the effect of confounding categorical variables: the African sub-region where the study was conducted (North, South, East, West and Central (see https://unstats.un.org/unsd/methodology/m49/), diagnostic technique (*S. mansoni*–Kato katz, Formol-ether or Bell, and *S. haematobium*–Filtration or sedimentation), and history of schistosomiasis treatment (No previous treatment in population, treatment within the past 5 years, treatment currently underway and history not given). Also included in the model was the interaction between SAC infection prevalence and treatment history, to determine if the relationship between the two prevalences differed according to the treatment history of the study population.

The relationship between PSAC and SAC was tested for (after allowing for the effects of the above mentioned confounding variables), using adjusted sums of squares to calculate the *F*-value. Adherence to the assumptions of parametric tests was confirmed by a series of residuals plots including histogram plots, residual vs. fits and residuals vs. plots. The significance value was p < 0.05.

The goodness of fit of the model to the data was recorded as the *r*-squared value, and the regression coefficient's 95% confidence intervals were calculated using the equation:

95% CI = $b \pm tc_{(n-2)} \ge SE$

Where;

b = regression coefficient

tc = Critical T = T value at 0.05 for n-2 degrees of freedom

n = study sample size.

SE = standard error of the coefficient

The 95% prediction band of the regression line were generated and plotted in GraphPad (Version 8). Statistical analyses were conducted with and without the data from the studies scoring low on the quality assessment (see <u>S3 Table</u> and <u>S4 Table</u>). This did not affect the result, and so the studies were included in the analysis.

Results

Literature search

The selection process is presented in Fig 1. Literature searches of online databases returned 3515 studies. 1263 duplications were identified and removed, leaving 2252 unique studies. After the screening and full text read, 68 studies and 143 datasets (89 for *S. mansoni* and 54 for *S. haematobium*) were used in the analysis.

Eligible studies

In total, 23 African countries had prevalence data which could be used in the analysis (Table 2) and these were spread across the continent (Fig 2).



Fig 1. Flowchart for search and selection of included studies.

https://doi.org/10.1371/journal.pntd.0008650.g001

	S. man	soni	S. haematobium		
Country	Publications	Data sets	Publications	Data sets	
Botswana	0	0	1	1	
Burundi	1	1	0	0	
Cameroon	0	0	2	5	
Cote d'Ivoire	4	7	5	6	
Egypt	5	8	2	3	
Ethiopia	4	8	1	1	
The Gambia	0	0	1	1	
Ghana	0	0	3	3	
Kenya	5	9	0	0	
Liberia	1	1	1	1	
Malawi	1	1	0	0	
Mali	1	1	1	1	
Niger	1	1	3	3	
Nigeria	1	1	7	7	
Senegal	2	2	2	2	
Sierra Leone	2	7	0	0	
Somalia	0	0	1	3	
South Africa	1	1	1	1	
Sudan	1	5	2	6	
Tanzania	1	1	3	2	
Uganda	4	5	0	0	
Zaire (DRC)	1	29	0	0	
Zimbabwe	1	1	4	8	

Table 2. Publications and data sets included in analysis, broken down by country of origin.

https://doi.org/10.1371/journal.pntd.0008650.t002

Schistosoma mansoni extracted data

There were 89 paired data points for *S. mansoni* extracted from 40 manuscripts with surveys between 1974 and 2018. <u>S1 Table</u>, shows characteristics of all studies used in the meta-analysis for *S. mansoni* infection in PSAC and SAC. Sample sizes ranged from 11 to 1122 in PSAC (mean = 145, median = 68). Sample sizes in SAC ranged from 12 to 2856 (mean = 333, median = 107). The prevalence of infection ranged from 0–100% in PSAC (mean = 25%, median = 15%, the interquartile range (IQR) = 33.1, standard deviation (SD) = 25.8) and 2.5–100% in SAC, (mean = 50%, median = 47%, IQR = 58.6, SD = 30.3).

For infection detection, the Kato-Katz technique was used in 86 of the 89 (96.6%) investigations, 2 (2.2%) used the formol-ether technique, and 1 (1.1%) used the Bell Method for egg detection in stool. Out of 89 data points, no treatment had previously taken place in 72 (80.9%) communities, 13 (14.6%) of the communities reported treatment currently underway, 1 (1.1%) study reported one round of MDA 2 years previously but no further treatment had taken place, and 2 communities (2.2%) did not report whether control strategies had taken place or not. For the *S. mansoni* analysis, 77 were Category 1 (data was from the same study) and 11 were from Category 2 (data was from two separate studies and adhere to the additional inclusion criteria).

Schistosoma haematobium extracted data

For *S. haematobium*, there were 54 paired data points from 39 manuscripts, and surveys occurred between 1962 and 2016. <u>S2 Table</u> shows characteristics of all studies that investigated





https://doi.org/10.1371/journal.pntd.0008650.g002

S. haematobium prevalence in PSAC and SAC. Sample sizes ranged from 11 to 1018 in PSAC (mean = 141, median = 78). Sample sizes of SAC ranged from 21 and 4326 (mean = 557, median = 215). The prevalence of infection ranged from 0–88% in PSAC (mean = 26.2%, median = 19.9%, IQR = 29.5, SD = 23.84) and 0–99% in SAC (mean = 51.5%, median = 54.2%, IQR = 36.7, SD = 27.9). Two diagnostic methods were included in the analysis, the filtration technique [20] was used in 36 of the 54 investigations, and the sedimentation technique was used in the remaining 18 studies.

No treatment had taken place prior to the study in 41 out of 54 (75.9%) of the data sets analysed, 8 out of 54 (14.8%) reported treatment currently underway, 1 (1.9%) study reported treatment of previously positive SAC with Niridazole, about 5 years previously, and 4 (7.4%) had no information on previous treatment status. For the *S. haematobium* analysis, 51 were Category 1, and 3 were from Category 2.

The sample sizes for the different independent variables are given in <u>Table 3</u>. Age range for PSAC across the published studies for both schistosome species was <1 up to 6 years, and for SAC, age range was 6–16 years. While for all studies PSAC sample sizes were lower than SAC

Variable	Categories	Schistosoma mansoni	Schistosoma haematobium
African Region	North	13	9
	South	1	2
	East	26	14
	West	20	24
	Central	29	5
Treatment History	No previous treatment	73	41
	Currently undergoing treatment	13	8
	MDA previously conducted in the area	1	1
	Treatment history information not provided	2	4
Diagnostic Method	Urine Filtration	N/A	36
	Sedimentation	N/A	18
	Kato Katz	87	N/A
	Formol Ether Technique	1	N/A
	Bell	1	N/A

Table 3. Summary of sample sizes for categories.

N/A = does not apply

https://doi.org/10.1371/journal.pntd.0008650.t003

sample sizes, the sample sizes recorded in the published studies for both species were correlated in PSAC and SAC i.e. those studies with low sample sizes for PSAC also had low sample sizes for SAC and the converse was true. Therefore, the regression analysis was weighted with the square root of the PSAC sample size.

Relationship in schistosome infection prevalence between preschool-aged children (PSAC) and school-aged children (SAC)

The Pearson correlation showed a linear relationship between PSAC infection prevalences and SAC infection prevalence in the same communities for both *S.mansoni* (r = 0.812, df = 1, p = <0.0001) and *S. haematobium* (r = 0.786, df = 1, p = <0.0001). This was subsequently analysed through linear regression.

The simple regression analysis was carried out without the effect of confounding variables, and was weighted by the square root of PSAC sample size. This showed a significant relationship between *S. mansoni* PSAC and SAC prevalences, with 65.9% of the variation in *S. mansoni* PSAC prevalence being explained by the model with SAC infection prevalence alone (95% CI = 54.7–77.1). Similarly, the regression analysis showed a significant linear relationship between *S. haemato-bium* PSAC and SAC prevalences with 61.8% of the variation in *S. haematobium* PSAC prevalence being explained by the model with SAC infection prevalence alone (95% CI = 46.4–77.2). The simple relationship between PSAC and SAC prevalences is shown in Fig 3A and 3B, and the results of the regression analysis are summarised in Table 4 (*S. mansoni*) and Table 5 (*S. haematobium*).

The WHO provides guidelines [4] for the optimal frequency of schistosome treatment of SAC, by assessing the prevalence of infection or visible haematuria in a community. We therefore coded the infection prevalence in PSAC using the guidelines for SAC (low, moderate and high) to indicate levels of infection in PSAC, which would be captured under those categories (see Fig 4). It can be seen that when SAC prevalence levels are low, there is a low infection prevalence in PSAC, whereas when infection levels are high in SAC, levels in PSAC are moderate to high. There are two instances where PSAC infection levels are higher than in SAC, only one published report for each schistosome species, and the fitted line in Fig 4 predicts that a 100% infection prevalence is reached earlier in SAC than PSAC.





https://doi.org/10.1371/journal.pntd.0008650.g003

The regression predicts PSAC prevalences for the WHO categories based on the SAC prevalences. For example, when SAC levels are low in *S. mansoni*, i.e. 0–9%, the PSAC levels are interpolated to be <1% (Fig 4A). Therefore based on the prediction, we determined the proportion of PSAC studies wrongly classified into a lower infection level, based on the SAC prevalences. Being, misclassified into a lower level is important from a public health perspective, as it means infection levels in PSAC will be underestimated.

For *S. mansoni*, the PSAC infection was underestimated in 4/89 study populations (4.5%). One study population (out of 4) had a PSAC prevalence that was classified in the low WHO category based on its SAC prevalence when it had a PSAC prevalence greater than 1% and therefore, should have been in the moderate category. This study reported the same prevalence 6.7% for both SAC and PSAC. Additionally, 3/42 (7.1%) study populations were classified as

Source			df	F-value	P- value
Regression			88 (1)	168.17	< 0.001
Prevalence in SAC			88 (1)	168.17	<0.001
Coefficients	Coef	SE Coef	95% CI Lower	95% CI Upper	P- value
Constant	-0.136	0.049	-0.265	-0.007	0.006
Prevalence in SAC	0.734	0.056	0.623	0.845	<0.001

Table 4. Analysis of variance and coefficients—S. mansoni, basic regression weighted by square root of PSACn.

Abbreviations: n-sample size, PSAC-preschool age children, SAC-School age children-square root arcsine transformed, df- degrees of freedom, Coef-coefficient, SE-Standard error.

Mathematical equation for the model: PSAC prevalence = A + bSAC prevalence

Where *A* is the constant coefficient and *b* is the Prevalence in SAC coefficient

Therefore: PSAC prevalence = -0.136 + 0.734 x SAC prevalence

https://doi.org/10.1371/journal.pntd.0008650.t004

moderate when their PSAC prevalences would have put them in high infection level. Of these 3 populations, 1 was undergoing treatment. For *S. haematobium*, in total 4/54 (7.4%) PSAC study populations were underestimated. One of the study populations (out of 4) had a PSAC prevalence that was classified in the low infection WHO category when it should have been in the moderate category. This study had a PSAC prevalence of 9% which was greater than the SAC prevalence of 7.9%. Further, 4/18 (22.2%) study populations were classified as moderate when their PSAC prevalences would have put them in high infection level. Of these 4 populations, 1 was undergoing treatment.

Effect of treatment history on the relationship between infection prevalence between preschool-aged children (PSAC) and school age children (SAC)

We conducted a regression analysis, allowing for the African sub-region where the study was conducted for both schistosome species, as well as the parasitology diagnostic method for *S*. *haematobium* only as there was not enough data in the diagnostic categories in the *S.mansoni* analysis to give a meaningful investigation. This was to determine if the relationship between PSAC and SAC prevalences varied significantly with the treatment history of the study community. The African sub-region was included in the analysis, as earlier descriptive statistics had indicated that there was heterogeneity in infection prevalences reported by studies from the different countries in Africa; this is confirmed by the significant effect of this variable on

Table 5.	Analysis of variance	e and coefficients-	-S. haematohium.	basic regression	weighted by s	auare root o	f PSACn.
Tuble 5. 1	and you of variance	c and coefficients	0. macmatooramy	ousic regression	i weighten by s	94410 10010	1 01101

Source			df	F-value	P- value	
Regression			54 (1)	84.28	< 0.001	
Prevalence in SAC			54 (1)	84.28	< 0.001	
Coefficients	Coef	SE Coef	95% CI Lower	95% CI Upper	P- value	
Constant	-0.065	0.062	-0.189	0.059	0.300	
Prevalence in SAC	0.680	0.074	0.531	0.828	< 0.001	

Abbreviations: n-sample size, PSAC-preschool age children, SAC-School age children-square root arcsine transformed, df- degrees of freedom, Coef-coefficient, SE-Standard error.

Mathematical equation for the model: PSAC prevalence = A + bSAC prevalence

Where A is the constant coefficient and b is the Prevalence in SAC coefficient

Therefore: PSAC prevalence = -0.065 + 0.680 x SAC prevalence

https://doi.org/10.1371/journal.pntd.0008650.t005



prevalence (%)	transformation	prevalence (%)	transformation
0 - 9%	0 - 0.31	< 1	0 - 0.091
10 - 49%	0.32 - 0.78	1-18%	0.099 - 0.436
50 - 100%	0.79 - 1.57	≥19	0.444 - 1.016



Fig 4. Scatterplots of PSAC vs SAC prevalence demarcated into the WHO prevalence classes for A) S. mansoni and B) S. haematobium. The classes are shown according to SAC prevalence levels showing the corresponding PSAC level on the Y-axis. Red = High, Orange = Moderate, Yellow = Low. The fitted line is from the linear regression analysis.

https://doi.org/10.1371/journal.pntd.0008650.g004

PSAC prevalence (see Table 6). We conducted the analysis with and without the studies from the Southern region which had the smallest sample sizes. Excluding these studies did not alter the results significantly, therefore we included all studies in the final analyses.

Source			df	F-value	P- value
Regression			88 (9)	25.63	< 0.001
Prevalence in SAC			88 (1)	163.15	< 0.001
African Regions (North, South, East, W	est and Central)		88 (4)	4.09	0.005
Treatment History			88 (2)	0.10	0.908
Prevalence in SAC * Treatment History	T		88 (2)	0.02	0.983
Coefficients	Coef	SE Coef	95% CI Lower	95% CI Upper	P- value
Constant	-0.137	0.065	-0.267	-0.008	0.038
Prevalence in SAC	0.7209	0.0564	0.6086	0.8333	< 0.001
African Regions:					0.005*
North	0.000	0.000	0.000	0.000	
South	-0.198	0.191	-0.578	0.182	0.303
East	-0.037	0.057	-0.151	0.077	0.519
West	0.055	0.063	-0.071	0.181	0.385
Central	0.132	0.064	0.004	0.260	0.044
Treatment History:					0.908*
No previous treatment	0.000	0.000	0.000	0.000	
Treatment underway	-0.060	0.153	-0.364	0.244	0.696
Unknown	0.46	2.65	-4.81	5.73	0.861
SAC *Treatment History:					0.983*
No previous treatment	0.000	0.000	0.000	0.000	
Treatment underway	-0.026	0.200	-0.425	0.373	0.897
Unknown	-1.08	7.71	-16.43	14.28	0.889

Table 6. Analysis of variance and coefficients-S. mansoni, regression weighted by square root of PSACn.

Abbreviations: n-sample size, PSAC-preschool age children, SAC-School age children-square root arcsine transformed, df- degrees of freedom, Coef-coefficient, SE-Standard error.

*Overall P-value from regression model

Mathematical equation for the model: PSAC prevalence = A + bSAC prevalence + cAfrican Region +dTreatment History + eSAC*Treatment History Where A is the constant coefficient and b,c,d and e are the variable coefficients.

https://doi.org/10.1371/journal.pntd.0008650.t006

Including the other variables and the interacting term in the model increased the *r*-squared value from 65.9% (95% CI = 54.7–77.1) to 74.5% (95% CI = 65.6–83.4) for *S. mansoni* and from 61.8% (95% CI = 46.4–77.2) to 70.3% (95% CI = 57.6–83.0) for *S. haematobium*. The correlation between PSAC and SAC prevalence was still significant. The results of this full model analysis are summarised in Tables <u>6</u> and <u>7</u>. This analysis showed that the relationship between PSAC and SAC prevalences was not significantly altered by treatment history (*S. mansoni*, *F* value = 0.02, df = 88, 2, p = 0.983, and *S haematobium*, *F* value = 0.36, df = 53, 2, p = 0.966).

It is worth noting that although statistical significance was not detected for the effect of treatment history on PSAC infection prevalence or for the interaction factor between SAC infection prevalence and treatment history, for both *S. mansoni* and *S. haematobium*, SAC infection levels in populations where schistosome treatment was underway (e.g. through MDA), consistently mapped to lower PSAC infection levels (see Fig 5).

Discussion

This is the first study investigating an association between the prevalence of schistosomiasis in African school-aged children (SAC) and preschool-aged children (PSAC). This study is driven by the current need to define strategies to include PSAC in schistosomiasis treatment programmes. The current WHO guidelines for treating PSAC is to treat children on a case-by-

Source			df	F-value	P- value
Regression			53(11)	10.20	< 0.0001
Prevalence in SAC			53(1)	57.42	< 0.0001
African Regions (North, South, East, V	West and Central)		53(4)	1.21	0.319
Treatment History			53(2)	0.33	0.721
Diagnostic Technique			53(1)	0.01	0.910
Prevalence in SAC * Treatment Histor	ry		53(2)	0.36	0.966
Coefficients	Coef	SE Coef	95% CI Lower	95% CI Upper	P- value
Constant	-0.095	0.102	-0.300	0.111	0.359
Prevalence in SAC	0.6776	0.894	0.497	0.8580	< 0.0001
African Regions:					0.319*
North	0.00	0.00	0.00	0.00	
South	0.007	0.177	-0.350	0.364	0.968
East	0.1080	0.098	-0.090	0.306	0.276
West	0.019	0.102	-0.187	0.224	0.855
Central	0.181	0.181	-0.093	0.454	0.189
Treatment history:					0.721*
No previous treatment	0.00	0.00	0.00	0.00	
Treatment underway	-0.074	0.183	-0.433	0.295	0.688
Unknown	-0.483	0.706	-1.908	0.941	0.497
Diagnostic technique:					0.910*
Filtration	0.00	0.00	0.00	0.00	
Sedimentation	0.007	0.056	-0.106	0.119	0.614
SAC * Treatment history:					0.966*
No previous treatment	0.00	0.00	0.00	0.00	
Treatment underway	-0.107	0.210	-0.529	0.319	0.614
Unknown	0.544	0.789	-1.047	2.135	0.494

Table 7. Analysis of variance and coefficients—S. haematobium, regression weighted by square root of PSACn.

Abbreviations: n-sample size, PSAC-preschool age children, SAC-School age children-square root arcsine transformed, df- degrees of freedom, Coef-coefficient, SE-Standard error.

*Overall P-value from regression model

Mathematical equation for the model: PSAC prevalence = A + bSAC prevalence + cAfrican Region +dTreatment History + eDiagnostic Technique + fSAC*Treatment history

Where *A* is the constant coefficient and *b*,*c*,*d*,*e* and *f* are the variable coefficients.

https://doi.org/10.1371/journal.pntd.0008650.t007

case basis upon diagnosis, with the current formulation of PZQ [13]. As the global community prepares for the deployment of the paediatric formulation of PZQ (currently undergoing Phase III clinical trials), and for eventual elimination of schistosomiasis, one of the main operational questions that remain unanswered is how to quantify infection in PSAC, and to identify areas/populations to target for treatment. PSAC are out of the school system and therefore not as easily accessible as SAC. A further operational question is, at what level of schistosome endemicity would it be practical to conduct an infection survey in this age group to diagnose and treat infections? We investigated the relationship between currently available data on SAC and PSAC within the same community to determine if data from SAC, who are the current target of control programmes could be used to predict infection levels in PSAC.

Previous studies have investigated using SAC as a proxy group for community-wide prevalence levels, showing that SAC prevalence is closely related to adult prevalence, and therefore this age group could be used as a proxy group [24, 25]. Our findings confirm that this approach



https://doi.org/10.1371/journal.pntd.0008650.g005

can be extended to quantify infection levels in PSAC. Our results show that PSAC infection levels are significantly correlated with SAC prevalence levels, and indicate that it is possible to extrapolate PSAC prevalence levels from known SAC prevalence and treat them accordingly. This finding is not surprising as SAC tend to use the same water sources as the rest of the community, and their contact with infective water is related to activities of older family members [26], so that when a high proportion of SAC are infected, the community risk of infection accumulates rapidly.

Several countries have embarked on preventative chemotherapy through mass drug administration (MDA). Thus, some of the data we analysed came from communities undergoing MDA programmes targeted at SAC, or in a few cases, targeting both SAC and adults, while others had not experienced MDA in the past 50 years; no current MDA programme includes PSAC. Therefore, we included the SAC, or SAC and adult's treatment history in our analyses. The current study indicated that the relationship between PSAC and SAC prevalences did not change significantly depending on treatment history of the study population. This finding may be due to the small sample sizes in the study because, albeit not significant in areas undergoing treatment, SAC levels translated to lower PSAC levels than in areas which had not experienced any treatment. In other words, to get the same levels of PSAC prevalence in areas that had undergone treatment, higher SAC prevalence levels were required than when compared to that in areas without treatment, for both S. mansoni and S. haematobium. In areas undergoing MDA, treatment of SAC has the potential of lowering levels of contamination of shared water sources with the infection, especially if treatment coverage is high and includes adults in the community; in this case, PSAC are exposed to lower levels of infection. In two examples where treatment was underway, one for S. mansoni and one for S. haematobium, infection prevalence in PSAC was greater than in SAC, presumably because the PSAC are excluded from the MDA programmes.

The maintenance of a relationship between PSAC and SAC prevalences during treatment, is relevant for most countries targeting SAC via preventative chemotherapy, and will need to include PSAC. This will become particularly prominent if the paediatric formulation of PZQ currently undergoing phase III clinical trials becomes available. This means that even data from national treatment impact assessment exercises that are mostly conducted by surveying SAC, are informative for developing strategies for treating PSAC. Thus, PSAC prevalence can be inferred from current levels of SAC prevalence data (as can be found in the Espen WHO database [27]), which can now be used to map prevalence levels of schistosomiasis in PSAC in Africa. Our results suggest that where prevalence of schistosomiasis infection is high in SAC, limited surveys could be used to validate infection in PSAC to justify preventive chemotherapy. This may obviate the need to conduct extensive surveys for schistosomiasis in PSAC. This is also in keeping with the WHO recommendation that given the financial and human resources required for surveys, existing epidemiological data including historical records should be used to generate baseline information for implementing a control strategy [6].

Unsurprisingly the relationship between SAC and PSAC indicates that infection levels in PSAC are zero or undetectable below a certain prevalence of SAC infection. This is due to both the low sensitivity of the widely used parasitological schistosomiasis diagnostic tests in detecting low intensity infections [28], as well as the infection transmission dynamics. Infection accumulates with age, with most children being exposed to infection within their first year of life, mainly as a result of increasing water contact and exposure patterns [29, 30]. While we can determine exposure to infection in the very young children serologically, infection detection by egg counts is less reliable because of the low levels of infection the children carry [29]. There is also the operational challenge of collecting stool and urine samples from the younger PSAC, who may not be toilet-trained; in this study PSAC age ranged from <1 year up to 6

years. Similarly, the number of samples collected would also affect the accuracy of diagnosis and not all studies indicated the number of urine or stool samples collected for the parasitology diagnosis.

Where we had sufficient data to investigate the influence of parasite egg collection methods, we found that for example with *S. haematobium*, the choice of egg collection methods, did not affect the PSAC infection prevalence. This is not surprising as both methods relied on enumerating parasite eggs excreted in urine, and thus suffer the same limitations of low sensitivity in people with low infection intensities [30].

Looking at methodological aspects, as with most meta-analyses and literature reviews, some publication bias can occur. Thus, we conducted a quality assessment exercise of the data published in the source reference. We then conducted the analyses both including and excluding the data from studies scoring low on the quality assessment, and as this did not alter the results, the data from all studies satisfying the inclusion criteria were kept in the analysis.

There were other possible analyses that would have been more informative, but sample sizes were too small for the data to be partitioned or analysed by those categories. For example, for both S. mansoni and S. haematobium, only one study reported previous, but no current MDA in the SAC. These both had their most recent MDA 2 years prior to the study quantifying SAC and PSAC infection, where the one examining S. haematobium used Niridazole instead of PZQ. The impact of MDA on the relationship between PSAC and SAC infection levels after cessation of MDA in SAC thus remains unexplored, but may likely not vary significantly due to infection rebound [31]. It would also have been informative to have the data for both SAC and PSAC in each age group, but this more granular data were not presented in the source publication. We attempted to capture some of the heterogeneity in the data by including factors such as the African sub-region where the data were collected, but other factors which could not be determined from the information presented in the publications might also account for the variation in the PSAC infection prevalence (the model explained 74.5% of the variation in S. mansoni and 70.3% in S. haematobium) This includes factors such as number of parasitology samples collected, number of slides prepared for diagnosis and age-related water contact. Nonetheless, the strong correlation between the two variables (i.e. PSAC and SAC prevalence) in these heterogeneous studies suggest that when quantifying infection in the same community with less heterogeneity, the SAC infection prevalence should be a more informative proxy for PSAC infection prevalence.

We determined the proportion of study populations classified into a lower infection level i.e. misclassified into a lower WHO classifications category based on the SAC levels. For *S. mansoni*, 4/89 (4.5%) of PSAC prevalences were underestimated and therefore misclassified, based on the interpolation of the regression equation. For *S. haematobium*, 4/54 (7.4%) of PSAC prevalences were misclassified. There are several reasons that could explain this, an example is treatment history. One of the *S. mansoni* studies reported that MDA was underway in SAC, and had similar prevalences in SAC and PSAC. The published source studies reported treatments via MDA but there was no indication if there had been any individual treatment provisions through health facilities. For *S. haematobium*, one study reported higher infection prevalence in PSAC than SAC, and the population had not had MDA targeted at the SAC. However, no explanation was given for this observation. The fact that even with the heterogeneities highlighted, over 90% of the studies were classified into the correct WHO classifications category based on the SAC levels from the regression model indicates that the SAC prevalences can be extended as a proxy for infection levels in PSAC as they are already being used for the adult population.

In conclusion, it has already been demonstrated that in schistosome endemic areas, PSAC are at high risk of schistosome infection and disease and therefore should be included in

treatment strategies. Here, we have demonstrated that schistosomiasis infection levels in PSAC are highly correlated to levels in SAC, giving the potential for using already existing data from SAC as a predictor of infection levels in PSAC, for both *S. mansoni* and *S. haematobium*. Critically, we have shown that the relationship between PSAC and SAC infection levels is informative during ongoing community or SAC-targeted treatment, as well as in communities that have not received schistosome treatment. With the increasing concern of the health impact of schistosomiasis in this PSAC, using already existing SAC data to inform PSAC chemotherapy could accelerate and facilitate the treatment of PSAC who are in urgent need of treatment. This will allow endemic countries to move closer towards the goal of eliminating schistosomiasis as a public health burden by 2025 [32].

Supporting information

S1 Checklist. PRISMA checklist. (DOCX)

S1 Table. Participant and methodology characteristics of studies included in meta-analysis containing *S. mansoni* data. (DOCX)

S2 Table. Participant and methodology characteristics of studies included in meta-analysis containing *S. haematobium* data. (DOCX)

S3 Table. Quality appraisal of articles containing prevalence data of *S. mansoni* infection in preschool and school age children in Africa. (DOCX)

S4 Table. Quality appraisal of articles containing prevalence data of *Schistosoma haemato-bium* infection in preschool and school age children in Africa. (DOCX)

Author Contributions

Conceptualization: Rivka M. Lim, Takafira Mduluza, Lester Chitsulo, Francisca Mutapi.

Data curation: Rivka M. Lim.

- Formal analysis: Rivka M. Lim, Mark E. J. Woolhouse, Margo Chase-Topping, Derick N. M. Osakunor, Francisca Mutapi.
- Writing original draft: Rivka M. Lim, Francisca Mutapi.
- Writing review & editing: Rivka M. Lim, Mark E. J. Woolhouse, Takafira Mduluza, Margo Chase-Topping, Derick N. M. Osakunor, Lester Chitsulo, Francisca Mutapi.

References

- de Vlas SJ, Danso-Appiah A, van der Werf MJ, Bosompem KM, Habbema JD. Quantitative evaluation of integrated schistosomiasis control: the example of passive case finding in Ghana. Tropical medicine & international health: TM & IH. 2004; 9(6):A16–21. https://doi.org/10.1111/j.1365-3156.2004.01260.x PMID: 15189470
- 2. World Health Organization. Schistosomiasis 2018 [cited 2018 22nd November]. Available from: http://www.who.int/en/news-room/fact-sheets/detail/schistosomiasis.

- Doenhoff M, Cioli D, Utzinger J. Praziquantel: mechanisms of action, resistance and new deriv...: Current Opinion in Infectious Diseases. 2018. https://doi.org/10.1097/QCO.0b013e328318978f PMID: 18978535
- 4. World Health Organization. Prevention and control of schistosomiasis and soil-transmitted helminthiasis. Geneva: 2002.
- Montresor A, Crompton DWT, Hall A, Bundy DAP, Savioli L. Guidelines for the evaluation of soil-transmitted helminthiasis and schistosomiasis at community level. 1998.
- 6. World Health Organization. Helminth control in school-age children: A guide for managers of control programmes: WHO Library Cataloguing; 2011.
- World Health Organization. Update on the global status of implementation of preventive chemotherapy (PC) 2018 [cited 2018 29th November 2018]. Available from: http://www.who.int/neglected_diseases/preventive_chemotherapy/PC_Update.pdf.
- Rudge J, Stothard R, Basanez M, Mgeni A, Khamis S, Khamis A, et al. Micro-epidemiology of urinary schistosomiasis in Zanzibar: Local risk factors associated with distribution of infections among schoolchildren and relevance for control. Acta tropica. 2008; 105(1):45–54. <u>https://doi.org/10.1016/j.</u> actatropica.2007.09.006 PMID: 17996207
- Utzinger J, N'Goran EK, Tanner M, Lengeler C. Simple anamnestic questions and recalled water-contact patterns for self-diagnosis of *Schistosoma mansoni* infection among schoolchildren in western Cote d'Ivoire. The American journal of tropical medicine and hygiene. 2000; 62(5):649–55. Epub 2001/04/06. https://doi.org/10.4269/ajtmh.2000.62.649 PMID: 11289679.
- Stothard JR, Sousa—Figueiredo J, Betson M, Bustinduy A, Reinhard-Rupp J. Schistosomiasis in African infants and preschool children: let them now be treated! Trends in parasitology. 2013; 29(4).
- Mutapi F. Changing policy and practice in the control of pediatric schistosomiasis. Pediatrics. 2015; 135 (3):536–44. Epub 2015/02/18. https://doi.org/10.1542/peds.2014-3189 PMID: 25687146.
- 12. World Health Organization. Ending the neglect to attain the sustainable development goals: a road map for neglected tropical diseases 2021–2030. World Health Organization, 2020.
- **13.** World Health Organization. Report of a meeting to review the results of studies on the treatment of schistosomiasis in preschool—age children. Geneva: 2010.
- Mutapi F, Rujeni N, Bourke C, Mitchell K, Appleby L, Nausch N, et al. Schistosoma haematobium treatment in 1–5 year old children: safety and efficacy of the antihelminthic drug praziquantel. PLoS neglected tropical diseases. 2011; 5(5):e1143. https://doi.org/10.1371/journal.pntd.0001143 PMID: 21610855.
- 15. Moher D, Liberati A, Tetzlaff J, Altman D, Group TP. Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS medicine. 2009; 6(7).
- Cooke A, Smith D, Booth A. Beyond PICO: The SPIDER Tool for Qualitative Evidence Synthesis. Qualitative Health Research. 2012; 22(10).
- 17. Ridley DS. The value of formol ether concentration of fecal cysts and ova. J Clin Pathol. 1956; 9:74–6. https://doi.org/10.1136/jcp.9.1.74 PMID: 13306797
- Bell DR. A new method for counting schistosoma mansoni eggs in faeces. Bull Wld Hlth Org. 1963; 29:525–30.
- Katz N, Chaves A, Pellegrino J. A simple device for quantitive stool thick smear technique in *schistoso-miasis mansoni*. Rev Instit Med Trop Sao Paulo. 1972; 14:397–400.
- **20.** Dazo BC, Biles JE. Two new field tecniques for detection and counting of *Schistosoma haematobium* eggs in urine samples, with evaluation of both methods. Bull Wld Hlth Org. 1974; 51:399–408.
- 21. World Health Organization. PCT Databank—schistosomiasis. 2019.
- 22. Flower A, McKenna JW, Upreti GJBm. Validity and reliability of GraphClick and DataThief III for data extraction. 2016; 40(3):396–413.
- 23. Sokal R, Rohlf F. Biometryl in the principle and practice of statistics in biological research. 3rdEdn. Freeman W. II and co. new York; 1995.
- Pereira A, Favre T, Galvad A, Beck L, Barbosa C, Pieri O. The prevalence of schistosomiasis in schoolaged children as an appropriate indicator of its prevalence in the community. Memorias do Instituto Oswaldo Cruz. 2010; 105(4):563–9. https://doi.org/10.1590/s0074-02762010000400036 PMID: 20721509
- **25.** Mwinzi PNM, Muchiri G, Wiegand RE, Omedo M, Abudho B, Karanja DMS, et al. Predictive value of school-aged children's schistosomiasis prevalence and egg intensity for other age groups in western kenya. American Journal of Tropical Medicine and Hygiene. 2015; 93(6):1311–7. <u>https://doi.org/10.4269/ajtmh.15-0467</u> PMID: 26416108

- Coulibaly JT, N'Gbesso YK, N'Guessan NA, Winkler MS, Utzinger J, N'Goran EK. Epidemiology of schistosomiasis in two high-risk communities of south Cote d'Ivoire with particular emphasis on preschool-aged children. The American journal of tropical medicine and hygiene. 2013; 89(1):32–41. Epub 2013/05/22. https://doi.org/10.4269/ajtmh.12-0346 PMID: 23690549; PubMed Central PMCID: PMC3748484.
- 27. World Health Organization. Expanded special project for elimination of neglected tropical diseases 2018 [cited 2019 01/03/2019]. Available from: http://espen.afro.who.int/.
- 28. Osakunor DN, Woolhouse M, Mutapi F. Paediatric schistosomiasis: What we know and what we need to know. PLoS neglected tropical diseases. 2018; 12(2).
- 29. Woolhouse ME, Mutapi F, Ndhlovu PD, Chandiwana SK, Hagan P. Exposure, infection and immune responses to *Schistosoma haematobium* in young children. Parasitology. 2000; 120 (Pt 1):37–44. Epub 2000/03/22. https://doi.org/10.1017/s0031182099005156 PMID: 10726264.
- Osakunor DM, T. Midzi N. Chase-Topping M. Mutsaka-Makuvaza M. Chimponda T. Eyoh E. Mduluza T Pfavayi L. Wami W. Amanfo S. Murray J. Tshuma C. Woolhouse M. Mutapi F. Dynamics of paediatric urogenital schistosome infection, morbidity and treatment: a longitudinal study among preschool children in Zimbabwe. 2018; 3(2). https://doi.org/10.1136/bmjgh-2017-000661 PMID: 29616147
- Mutapi F, Maizels R, Fenwick A, Woolhouse M. Human schistosomiasis in the post mass drug administration era. Lancet Infect Dis. 2017; 17(2):e42–e8. <u>https://doi.org/10.1016/S1473-3099(16)30475-3</u> PMID: 27988094.
- **32.** World Health Organization. Schistosomiasis: progress report 2001–2011, strategic plan 2012–2020. World Health Organization, 2013.